Author's response to reviews

Title: The effectiveness of reducing the daily dose of finasteride in men with benign prostatic hyperplasia

Authors:

Michael J Sullivan MD, FACP (drslullivan@ijm.net)
Jack Geller MD, MACP (geller@san.rr.com)

Version: 2 Date: 6 Jan 2002

Title: The effectiveness of reducing the daily dose of finasteride in men with benign prostatic hyperplasia

Reviewer's suggestions have been incorporated into the article.

With respect to Dr. Wolfsen's comment on paragraph 4, line 1 this reads correctly, Day 1 was after one year of finasteride 5 mg daily. The One year, see table, is after one year of 2.5 mg finasteride daily.

Dr. Wolfsen points out correctly that total symptom scores were not different between 5 mg and 1 mg finasteide but obstructive symptoms were significantly improved with 1 mg finasteride and this is clarified in the discussion.